Author’s response to reviews

Title: Day hospital Mentalization-Based Treatment versus intensive outpatient Mentalization-Based Treatment for patients with severe borderline personality disorder: Protocol of a multicentre randomized clinical trial

Authors:
- Elisabeth M.P. Laurenssen (annelies.laurenssen@deviersprong.nl)
- Maaike L. Smits (maaike.smits@deviersprong.nl)
- Dawn L. Bales (dawn.bales@deviersprong.nl)
- Dine J. Feenstra (dineke.feenstra@deviersprong.nl)
- Hester V. Eeren (h.vaneeren1@erasmusmc.nl)
- Marc J. Noom (noom.m@zaansmc.nl)
- Maartje A. Köster (maartje.koster@npsai.nl)
- Zwaan Lucas (z.a.lucas@groningen.psyq.nl)
- Reinier Timman (r.timman@erasmusmc.nl)
- Jack J.M. Dekker (jack.dekker@arkin.nl)
- Patrick Luyten (patrick.luyten@ppw.kuleuven.be)
- Jan J.V. Busschbach (jan.busschbach@deviersprong.nl)
- Roel Verheul (roel.verheul@deviersprong.nl)

Version: 2 Date: 15 October 2014

Author’s response to reviews: see over
Dear editor,

Hereby, we submit the revised version of “Day hospital Mentalization-Based treatment versus Intensive Outpatient Mentalization-Based Treatment for patients with severe borderline personality disorder: Protocol of a multi-centre head-to-head randomized clinical trial”. This paper presents the background, aims, and design of an RCT study comparing two dosages of MBT for patients with BPD.

As you requested, we added the date of registration in the Dutch Trial Register on page 3. We added “Registered 16 April 2010.”

Yours sincerely, also on behalf of my co-authors,

Elisabeth

Viersprong Institute for Studies on Personality Disorders (VISPD)
P.O. Box 7
4660 AA Halsteren
The Netherlands
Annelies.laurenssen@deviersprong.nl